\begin{table}%t2
%\centering
\par
\caption{\label{Table_section}The details of all the analysed pulsars. The classification of the pulsar in the second column, where "Coh'' is a coherent drifter, "Dif'' and "Dif$^\ast$'' are diffuse drifters with or without drift features which are clearly separated from the alias borders and "Lon'' are pulsars showing longitude stationary subpulse modulation. The next columns are the pulse period, its dimensionless time derivative, the number of pulses in the observation, the signal to noise ratio, the minimum in the longitude resolved modulation index, the minimum detectable modulation index, the horizontal and vertical driftband separation and the figure number.} 
\begin{tabular}{ll|rc|rr|rrcc|l}
\hline\hline
Pulsar & Class  & $P_0$ (s) & $\dot P$ & Pulses & {\it S/N}\hspace{1.0mm} &  $m$\hspace{5mm} & $m_{\rm thresh}$ & $P_2$  (deg)& $P_3$  $(P_0)$&  Figure\\
\hline
B0011+47 &    & 1.2407 & $ 5.6\times10^{-16}$ & 1564 & 105  & $0.7\pm0.1$ & 0.31 & &   &  \ref{B0011+47}\\
B0031$-$07 & Dif  & 0.9430 & $ 4.1\times10^{-16}$ & 1204 & 54  & $1.2\pm0.1$ & 0.51 & $ -40^{\rm +2}_{-50}$ & $ 8.3\pm0.3$ &  \ref{B0031-07}\\
B0037+56 & Dif$^\ast$   & 1.1182 & $ 2.9\times10^{-15}$ & 2049 & 131  & $0.80\pm0.06$ & 0.22 & $ 7^{\rm +10}_{-2}$ & $22\pm3$ &  \ref{B0037+56}\\
B0052+51 & Dif  & 2.1152 & $ 9.5\times10^{-15}$ & 1112 & 112  & $0.92\pm0.06$ & 0.18 & $ -75^{\rm +50}_{-50}$ & $4\pm2$ &  \ref{B0052+51}\\
 &  &    & &  & & &  & $ 30^{\rm +70}_{-7}$ & $5\pm1$ &  \\
B0105+65 &    & 1.2837 & $ 1.3\times10^{-14}$ & 1346 & 71  & $0.43\pm0.04$ & 0.33 & &   &  \ref{B0105+65}\\
J0134$-$2937 &    & 0.1370 & $ 7.8\times10^{-17}$ & 4096 & 35  &  & 0.67 & &   &  \ref{J0134-2937}\\
B0136+57 & Dif$^\ast$  & 0.2725 & $ 1.1\times10^{-14}$ & 3432 & 450  & $0.49\pm0.01$ & 0.09 & $ -70^{\rm +15}_{-80}$ & $ 6.5\pm0.2$ &  \ref{B0136+57}\\
B0138+59 & Dif$^\ast$  & 1.2229 & $ 3.9\times10^{-16}$ & 2560 & 381  & $0.47\pm0.01$ & 0.09 & $ 300^{\rm +140}_{-110}$ & $15\pm4$ &  \ref{B0138+59}\\
B0144+59 &   & 0.1963 & $ 2.6\times10^{-16}$ & 13534 & 151  & $1.20\pm0.07$ & 0.36 &   &  &  \ref{B0144+59}\\
B0148$-$06 & Coh   & 1.4647 & $ 4.4\times10^{-16}$ & 1400 & 135  & $0.59\pm0.04$ & 0.22 & $ -12.5^{\rm +0.4}_{-1.9}$ & $ 14.2\pm0.2$ &  \ref{B0148-06}\\
 &  &    & &  & & &  & $ -45^{\rm +7}_{-20}$ & $ 14.7\pm0.3$ &  \\
B0149$-$16 & Coh   & 0.8327 & $ 1.3\times10^{-15}$ & 1024 & 62  & $0.67\pm0.05$ & 0.42 & $ -9^{\rm +12}_{-1}$ &$ 5.8\pm0.5$  &  \ref{B0149-16}\\
B0154+61 &    & 2.3517 & $ 1.9\times10^{-13}$ & 769 & 138  & $1.2\pm0.1$ & 0.15 &  &  &  \ref{B0154+61}\\
B0301+19 & Dif$^\ast$  & 1.3876 & $ 1.3\times10^{-15}$ & 1263 & 129  & $0.84\pm0.04$ & 0.23 & $ -35^{\rm +7}_{-55}$ & $ 5.2\pm0.3$ &  \ref{B0301+19}\\
B0320+39 & Coh  & 3.0321 & $ 6.4\times10^{-16}$ & 7169 & 443  & $0.21\pm0.01$ & 0.14 & $ 18^{\rm +5}_{-3}$ & $ 8.4\pm0.1$ &  \ref{B0320+39}\\
B0329+54 & Dif$^\ast$  & 0.7145 & $ 2.0\times10^{-15}$ & 19969 & 19064  & $0.423\pm0.001$ & 0.01 & $ -200^{\rm +30}_{-150}$ & $3\pm2$ &  \ref{B0329+54}\\
 &  &    & &  & & &  & $ 70^{\rm +13}_{-6}$ & $5\pm1$ &  \\
B0353+52 &    & 0.1970 & $ 4.8\times10^{-16}$ & 8866 & 117  & $0.63\pm0.06$ & 0.28 &  &  &  \ref{B0353+52}\\
B0355+54 &    & 0.1564 & $ 4.4\times10^{-15}$ & 11264 & 1133  & $0.77\pm0.01$ & 0.07 & &   &  \ref{B0355+54}\\
B0402+61 & Lon   & 0.5946 & $ 5.6\times10^{-15}$ & 12150 & 156  & $0.7\pm0.1$ & 0.37 & &   &  \ref{B0402+61}\\
 &  &    & &  & & &  & ? & $23\pm4$ &  \\
B0450+55 & Dif$^\ast$  & 0.3407 & $ 2.4\times10^{-15}$ & 2664 & 1095  & $0.49\pm0.01$ & 0.06 & $ 200^{\rm +70}_{-120}$ & $ 20\pm10$ &  \ref{B0450+55}\\
 &  &    & &  & & &  & $ -70^{\rm +8}_{-25}$ & $10\pm1$ &  \\
B0450$-$18 &    & 0.5489 & $ 5.8\times10^{-15}$ & 1537 & 362  & $0.31\pm0.01$ & 0.13 &  &  &  \ref{B0450-18}\\
B0458+46 &    & 0.6386 & $ 5.6\times10^{-15}$ & 1440 & 92  & $0.46\pm0.07$ & 0.43 & &   &  \ref{B0458+46}\\
B0523+11 & Dif  & 0.3544 & $ 7.4\times10^{-17}$ & 14849 & 206  & $0.56\pm0.05$ & 0.34 & $ 30^{\rm +10}_{-11}$ & $ 3.0\pm0.2$ &  \ref{B0523+11}\\
B0525+21 & Dif$^\ast$  & 3.7455 & $ 4.0\times10^{-14}$ & 1029 & 429  & $1.15\pm0.02$ & 0.05 & $ -20^{\rm +2}_{-9}$ & $ 3.8\pm0.7$ &  \ref{B0525+21}\\
 &  &    & &  & & &  & $ 50^{\rm +55}_{-10}$ & $ 3.7\pm0.4$ &  \\
B0531+21 &    & 0.0331 & $ 4.2\times10^{-13}$ & 178951 & 294  & $5.31\pm0.04$ & 0.28 &   & &  \ref{B0531+21}\\
B0540+23 &    & 0.2460 & $ 1.5\times10^{-14}$ & 3073 & 773  & $1.29\pm0.01$ & 0.04 & &   &  \ref{B0540+23}\\
B0559$-$05 &    & 0.3960 & $ 1.3\times10^{-15}$ & 2049 & 73  & $0.63\pm0.07$ & 0.46 & &   &  \ref{B0559-05}\\
B0609+37 & Coh  & 0.2980 & $ 5.9\times10^{-17}$ & 2561 & 139  & $0.56\pm0.03$ & 0.22 & $ -20^{\rm +4}_{-18}$ & $ 32\pm4$ &  \ref{B0609+37}\\
B0611+22 &    & 0.3350 & $ 5.9\times10^{-14}$ & 2560 & 139  & $0.59\pm0.02$ & 0.26 & &   &  \ref{B0611+22}\\
B0621$-$04 & Coh  & 1.0391 & $ 8.3\times10^{-16}$ & 1536 & 38  & $0.85\pm0.09$ & 0.47 & $ 25^{\rm +14}_{-16}$ & $ 2.055\pm0.001$ &  \ref{B0621-04}\\
B0626+24 &    & 0.4766 & $ 2.0\times10^{-15}$ & 1025 & 248  & $0.29\pm0.01$ & 0.14 & &   &  \ref{B0626+24}\\
B0628$-$28 & Dif$^\ast$  & 1.2444 & $ 7.1\times10^{-15}$ & 4267 & 1064  & $0.59\pm0.02$ & 0.07 & $ 30^{\rm +80}_{-6}$ & $ 7\pm1$ &  \ref{B0628-28}\\
B0656+14 &    & 0.3849 & $ 5.5\times10^{-14}$ & 2065 & 292  & $1.24\pm0.03$ & 0.10 & &   &  \ref{B0656+14}\\
%B0727$-$18 & Dif$^\ast$   & 0.5102 & $ 1.9\times10^{-14}$ & 3632 & 97  & $0.89\pm0.04$ & 0.34 & $ -160^{\rm +60}_{-110}$ & $ 22$&$\pm5$ &  \ref{B0727-18}\\
B0740$-$28 &    & 0.1668 & $ 1.7\times10^{-14}$ & 3585 & 380  & $0.27\pm0.02$ & 0.12 &   & &  \ref{B0740-28}\\
B0751+32 &  Dif$^\ast$  & 1.4423 & $ 1.1\times10^{-15}$ & 774 & 60  & $0.89\pm0.08$ & 0.31 & $ -30^{\rm +15}_{-300}$ & $ 5\pm5$ &  \ref{B0751+32}\\
B0756$-$15 &    & 0.6823 & $ 1.6\times10^{-15}$ & 1224 & 61  & $0.62\pm0.07$ & 0.42 & &   &  \ref{B0756-15}\\
B0809+74 & Coh  & 1.2922 & $ 1.7\times10^{-16}$ & 13092 & 2635  & $0.496\pm0.001$ & 0.04 & $ -16^{\rm +1}_{-16}$ & $ 11.1\pm0.1$ &  \ref{B0809+74}\\
B0818$-$13 & Coh  & 1.2381 & $ 2.1\times10^{-15}$ & 2251 & 345  & $0.34\pm0.01$ & 0.12 & $ -6.5^{\rm +0.2}_{-0.7}$ & $ 4.7\pm0.2$ &  \ref{B0818-13}\\
B0820+02 &    & 0.8649 & $ 1.0\times10^{-16}$ & 971 & 42  & $0.53\pm0.08$ & 0.41 & &   &  \ref{B0820+02}\\
B0823+26 & Dif$^\ast$  & 0.5307 & $ 1.7\times10^{-15}$ & 1596 & 2635  & $0.967\pm0.001$ & 0.01 & $ 55^{\rm +40}_{-7}$ & $7\pm2$ &  \ref{B0823+26}\\
B0834+06 & Dif$^\ast$  & 1.2738 & $ 6.8\times10^{-15}$ & 1032 & 199  & $0.47\pm0.05$ & 0.14 & $ 20^{\rm +55}_{-9}$ & $ 2.2\pm0.2$ &  \ref{B0834+06}\\
 &  &    & &  & & &  & $ 40^{\rm +140}_{-4}$ & $ 2.1\pm0.2$ &  \\
B0906$-$17 &    & 0.4016 & $ 6.7\times10^{-16}$ & 2115 & 46  & $0.8\pm0.2$ & 0.62 & &   &  \ref{B0906-17}\\
B0919+06 & Dif$^\ast$  & 0.4306 & $ 1.4\times10^{-14}$ & 8255 & 823  & $0.620\pm0.002$ & 0.04 & $ -150^{\rm +13}_{-75}$ & $ 4.0\pm0.8$ &  \ref{B0919+06}\\
B0950+08 &    & 0.2531 & $ 2.3\times10^{-16}$ & 1311 & 422  & $1.08\pm0.02$ & 0.27 & &  &   \ref{B0950+08}\\
J1022+1001 &    & 0.0165 & $ 4.3\times10^{-20}$ & 211891 & 251  & $0.60\pm0.07$ & 0.50 &  & &   \ref{J1022+1001}\\
B1039$-$19 & Dif  & 1.3864 & $ 9.4\times10^{-16}$ & 929 & 160  & $0.51\pm0.08$ & 0.18 & $ 25^{\rm +9}_{-4}$ & $ 4.3\pm0.3$ &  \ref{B1039-19}\\
 &  &    & &  & & &  & $ 9.0^{\rm +3.5}_{-0.1}$ & $ 4.3\pm0.1$ &  \\
B1112+50 &    & 1.6564 & $ 2.5\times10^{-15}$ & 1599 & 147  & $1.5\pm0.2$ & 0.27 & &  &   \ref{B1112+50}\\
B1133+16 & Dif$^\ast$  & 1.1879 & $ 3.7\times10^{-15}$ & 1514 & 1156  & $1.4\pm0.1$ & 0.04 & $ 200^{\rm +55}_{-90}$ & $3\pm2$ &  \ref{B1133+16}\\
 &  &    & &  & & &  & $ 130^{\rm +120}_{-15}$ & $3\pm1$ &  \\
B1237+25 & Dif$^\ast$  & 1.3824 & $ 9.6\times10^{-16}$ & 1265 & 2121  & $0.480\pm0.002$ & 0.04 & $ -20^{\rm +2}_{-3}$ & $ 2.7\pm0.1$ &  \ref{B1237+25}\\
 &  &    & &  & & &  & $ 16^{\rm +1}_{-1}$ & $ 2.7\pm0.1$ &  \\
B1254$-$10 &    & 0.6173 & $ 3.6\times10^{-16}$ & 1389 & 14  &  & 1.58 & &  &   \ref{B1254-10}\\
B1508+55 & Dif$^\ast$  & 0.7397 & $ 5.0\times10^{-15}$ & 4808 & 568  & $0.52\pm0.01$ & 0.09 & $ -160^{\rm +40}_{-80}$ & $ 5\pm5$ &  \ref{B1508+55}\\
J1518+4904 & Dif   & 0.0409 & $ 2.7\times10^{-20}$ & 31723 & 287  & $0.36\pm0.01$ & 0.13 & $ 55^{\rm +25}_{-4}$ & $ 2.6\pm0.1$ &  \ref{J1518+4904}\\
B1540$-$06 & Coh  & 0.7091 & $ 8.8\times10^{-16}$ & 5121 & 293  & $0.27\pm0.03$ & 0.14 & $ 24^{\rm +9}_{-4}$ & $ 3.03\pm0.02$ &  \ref{B1540-06}\\
 &  &    & &  & & &  & $ -16^{\rm +12}_{-45}$ & $ 3.02\pm0.02$ &  \\
B1541+09 &    & 0.7484 & $ 4.3\times10^{-16}$ & 1126 & 93  & $0.47\pm0.07$ & 0.41 & &   &  \ref{B1541+09}\\
B1600$-$27 &    & 0.7783 & $ 3.0\times10^{-15}$ & 1077 & 34  & $0.7\pm0.1$ & 0.71 & &   &  \ref{B1600-27}\\
B1604$-$00 & Dif$^\ast$  & 0.4218 & $ 3.1\times10^{-16}$ & 2012 & 1958  & $0.61\pm0.07$ & 0.02 & $ 55^{\rm +100}_{-3}$ & $ 3.4\pm0.5$ &  \ref{B1604-00}\\
 &  &    & &  & & &  & $ 70^{\rm +35}_{-5}$ & $ 3.1\pm0.5$ &  \\
B1612+07 &    & 1.2068 & $ 2.4\times10^{-15}$ & 1926 & 56  & $0.68\pm0.05$ & 0.45 & &   &  \ref{B1612+07}\\
B1642$-$03 & Dif$^\ast$  & 0.3877 & $ 1.8\times10^{-15}$ & 3118 & 656  & $0.337\pm0.002$ & 0.06 & $ 60^{\rm +9}_{-4}$ & $ 15\pm1$ &  \ref{B1642-03}\\
B1649$-$23 &    & 1.7037 & $ 3.2\times10^{-15}$ & 1025 & 38  & $1.0\pm0.2$ & 0.61 & & &    \ref{B1649-23}\\
J1650$-$1654 & Coh  & 1.7496 & $ 3.2\times10^{-15}$ & 1002 & 55  & $0.7\pm0.1$ & 0.49 & $ 9^{\rm +7}_{-1}$ & $ 2.59\pm0.05$ &  \ref{J1650-1654}\\
B1702$-$19 & Coh  & 0.2990 & $ 4.1\times10^{-15}$ & 11071 & 591  & $0.34\pm0.01$ & 0.13 & $ -80^{\rm +6}_{-70}$ & $ 11.0\pm0.4$ &  \ref{B1702-19}\\
 &  &    & &  & & &  & ? & $ 11.0\pm0.4$ &  \\
B1706$-$16 &    & 0.6531 & $ 6.3\times10^{-15}$ & 1309 & 377  & $0.71\pm0.06$ & 0.12 & &  &   \ref{B1706-16}\\
J1713+0747 &    & 0.0046 & $ 8.5\times10^{-21}$ & 791257 & 181  & $0.60\pm0.04$ & 0.40 &  & &   \ref{J1713+0747}\\
B1717$-$16 &    & 1.5656 & $ 5.8\times10^{-15}$ & 1120 & 104  & $0.65\pm0.08$ & 0.22 & &  &   \ref{B1717-16}\\
B1717$-$29 & Coh  & 0.6204 & $ 7.5\times10^{-16}$ & 1794 & 17  & $1.0\pm0.2$ & 1.24 & $ -9.6^{\rm +3}_{-0.6}$ & $ 2.45\pm0.02$ &  \ref{B1717-29}\\
B1730$-$22 &    & 0.8717 & $ 4.3\times10^{-17}$ & 880 & 76  & $0.41\pm0.04$ & 0.32 & &  &   \ref{B1730-22}\\
J1730$-$2304 &    & 0.0081 & $ 2.0\times10^{-20}$ & 103617 & 76  &  & 1.52 & & &    \ref{J1730-2304}\\
B1732$-$07 &    & 0.4193 & $ 1.2\times10^{-15}$ & 1826 & 79  & $0.43\pm0.04$ & 0.33 & &  &   \ref{B1732-07}\\
B1736$-$29 &    & 0.3229 & $ 7.9\times10^{-15}$ & 2606 & 26  & $1.0\pm0.2$ & 0.89 & & &   \ref{B1736-29}\\
B1737+13 &    & 0.8031 & $ 1.5\times10^{-15}$ & 1037 & 30  & $0.9\pm0.2$ & 0.73 & & &    \ref{B1737+13}\\
B1738$-$08 & Dif$^\ast$  & 2.0431 & $ 2.3\times10^{-15}$ & 859 & 153  & $0.86\pm0.04$ & 0.27 & $ 60^{\rm +20}_{-7}$ & $ 5.2\pm0.1$ &  \ref{B1738-08}\\
 &  &    & &  & & &  & $ 9.5^{\rm +4}_{-0.5}$ & $  4.4\pm0.3$ &  \\
B1744$-$24A &    & 0.0116 & $ -3.4\times10^{-20}$ & 309244 & 34  &  & 2.21 & & &    \ref{B1744-24A}\\
B1745$-$12 &    & 0.3941 & $ 1.2\times10^{-15}$ & 2054 & 75  & $0.49\pm0.08$ & 0.49 & & &    \ref{B1745-12}\\
B1749$-$28 &   & 0.5626 & $ 8.1\times10^{-15}$ & 1285 & 898  & $0.540\pm0.004$ & 0.04 & & &    \ref{B1749-28}\\
B1753+52 & Dif  & 2.3914 & $ 1.6\times10^{-15}$ & 784 & 60  & $0.73\pm0.09$ & 0.50 & $ 100^{\rm +30}_{-60}$ & $ 11\pm5$ &  \ref{B1753+52}\\
 &  &    & &  & & &  & $ 9.7^{\rm +3}_{-0.7}$ & $ 6.3\pm0.4$ &  \\
B1754$-$24 &    & 0.2341 & $ 1.3\times10^{-14}$ & 3589 & 113  & $0.58\pm0.06$ & 0.27 & & &    \ref{B1754-24}\\
B1756$-$22 &    & 0.4610 & $ 1.1\times10^{-14}$ & 1561 & 103  & $0.46\pm0.07$ & 0.19 & & &    \ref{B1756-22}\\
J1757$-$2223 &    & 0.1853 & $ 7.8\times10^{-16}$ & 4629 & 9  &  & 2.04 & &   &  \ref{J1757-2223}\\
B1758$-$23 &    & 0.4158 & $ 1.1\times10^{-13}$ & 2014 & 87  &  & 0.48 & & &    \ref{B1758-23}\\
B1758$-$29 &    & 1.0819 & $ 3.3\times10^{-15}$ & 1067 & 19  & $0.9\pm0.2$ & 1.10 & & &    \ref{B1758-29}\\
B1800$-$21 &    & 0.1336 & $ 1.3\times10^{-13}$ & 6210 & 134  & $0.46\pm0.04$ & 0.27 & &  &   \ref{B1800-21}\\
B1804$-$08 &    & 0.1637 & $ 2.9\times10^{-17}$ & 5241 & 414  & $0.30\pm0.03$ & 0.12 & &  &   \ref{B1804-08}\\
B1805$-$20 &    & 0.9184 & $ 1.7\times10^{-14}$ & 1056 & 102  & $0.51\pm0.04$ & 0.23 & &  &   \ref{B1805-20}\\
J1808$-$0813 &    & 0.8760 & $ 1.2\times10^{-15}$ & 974 & 73  & $0.52\pm0.09$ & 0.41 & &  &   \ref{J1808-0813}\\
B1811+40 &    & 0.9311 & $ 2.5\times10^{-15}$ & 7200 & 109  & $0.56\pm0.1$ & 0.53 & &  &   \ref{B1811+40}\\
J1812$-$2102 &    & 1.2234 & $ 2.4\times10^{-14}$ & 1435 & 63  &  & 0.65 & &  &   \ref{J1812-2102}\\
B1813$-$17 &    & 0.7823 & $ 7.3\times10^{-15}$ & 1096 & 50  & $0.7\pm0.1$ & 0.58 & & &    \ref{B1813-17}\\
B1815$-$14 &    & 0.2915 & $ 2.0\times10^{-15}$ & 2579 & 175  & $0.27\pm0.03$ & 0.18 & &   &  \ref{B1815-14}\\
B1817$-$13 &    & 0.9215 & $ 4.5\times10^{-15}$ & 1031 & 46  & $0.65\pm0.06$ & 0.41 & &   &  \ref{B1817-13}\\
B1818$-$04 &   & 0.5981 & $ 6.3\times10^{-15}$ & 7681 & 353  & $0.40\pm0.04$ & 0.16 & &   &  \ref{B1818-04}\\
B1819$-$22 & Dif  & 1.8743 & $ 1.4\times10^{-15}$ & 1096 & 258  & $0.76\pm0.01$ & 0.10 & $ -9.1^{\rm +0.1}_{-1.0}$ & $ 9.8\pm0.6$ &  \ref{B1819-22}\\
B1820$-$11 &    & 0.2798 & $ 1.4\times10^{-15}$ & 3021 & 97  & $0.61\pm0.04$ & 0.35 & & &    \ref{B1820-11}\\
B1821+05 &    & 0.7529 & $ 2.3\times10^{-16}$ & 1080 & 24  &  & 0.99 & & &    \ref{B1821+05}\\
B1821$-$11 &    & 0.4358 & $ 3.6\times10^{-15}$ & 1568 & 59  &  & 0.37 & &   &  \ref{B1821-11}\\
B1821$-$19 &    & 0.1893 & $ 5.2\times10^{-15}$ & 4105 & 222  & $0.632\pm0.004$ & 0.08 & &  &   \ref{B1821-19}\\
B1822$-$09 & Dif$^\ast$  & 0.7690 & $ 5.2\times10^{-14}$ & 1180 & 409  & $0.672\pm0.005$ & 0.24 & $ 50^{\rm +110}_{-15}$ & $ 8\pm4$ &  \ref{B1822-09}\\
B1822$-$14 &    & 0.2792 & $ 2.3\times10^{-14}$ & 3061 & 79  & $1.8\pm0.2$ & 0.48 & & &    \ref{B1822-14}\\
B1826$-$17 &    & 0.3071 & $ 5.6\times10^{-15}$ & 2764 & 361  & $0.59\pm0.01$ & 0.11 & &  &   \ref{B1826-17}\\
J1828$-$1101 &    & 0.0721 & $ 1.5\times10^{-14}$ & 24167 & 64  &  & 0.98 & &  &   \ref{J1828-1101}\\
B1829$-$08 &    & 0.6473 & $ 6.4\times10^{-14}$ & 1311 & 132  & $0.56\pm0.09$ & 0.22 & &  &   \ref{B1829-08}\\
B1830$-$08 &    & 0.0853 & $ 9.2\times10^{-15}$ & 9412 & 124  & $1.4\pm0.2$ & 0.22 & & &    \ref{B1830-08}\\
J1830$-$1135 & Dif$^\ast$  & 6.2216 & $ 4.8\times10^{-14}$ & 1004 & 140  & $1.1\pm0.1$ & 0.21 & $ 25^{\rm +25}_{-7}$ & $ 2.1\pm0.2$ &  \ref{J1830-1135}\\
B1831$-$03 &    & 0.6867 & $ 4.2\times10^{-14}$ & 1236 & 88  & $0.45\pm0.06$ & 0.26 & & &    \ref{B1831-03}\\
B1831$-$04 &    & 0.2901 & $ 7.2\times10^{-17}$ & 11255 & 114  & $0.41\pm0.05$ & 0.36 & &  &   \ref{B1831-04}\\
B1832$-$06 &    & 0.3058 & $ 4.0\times10^{-14}$ & 4557 & 33  &  & 0.98 & & &    \ref{B1832-06}\\
B1834$-$04 &    & 0.3542 & $ 1.7\times10^{-15}$ & 4114 & 73  & $0.57\pm0.05$ & 0.41 & & &    \ref{B1834-04}\\
B1834$-$10 &    & 0.5627 & $ 1.2\times10^{-14}$ & 1286 & 151  & $0.32\pm0.01$ & 0.12 & & &    \ref{B1834-10}\\
J1835$-$1106 &    & 0.1659 & $ 2.1\times10^{-14}$ & 5125 & 108  & $0.89\pm0.04$ & 0.42 & & &    \ref{J1835-1106}\\
B1839+09 &    & 0.3813 & $ 1.1\times10^{-15}$ & 2056 & 79  & $0.48\pm0.06$ & 0.41 & & &    \ref{B1839+09}\\
B1839+56 &    & 1.6529 & $ 1.5\times10^{-15}$ & 1031 & 53  & $1.07\pm0.06$ & 0.44 & & &    \ref{B1839+56}\\
B1839$-$04 & Coh  & 1.8399 & $ 5.1\times10^{-16}$ & 1025 & 179  & $0.49\pm0.04$ & 0.15 & $ -35^{\rm +5}_{-15}$ & $ 12.4\pm0.3$ &  \ref{B1839-04}\\
 &  &    & &  & & &  & $ 120^{\rm +20}_{-20}$ & $ 12.4\pm0.3$ &  \\
J1839$-$0643 &    & 0.4495 & $ 3.6\times10^{-15}$ & 3590 & 50  & $0.51\pm0.08$ & 0.47 & & &    \ref{J1839-0643}\\
B1841$-$04 & Coh  & 0.9910 & $ 3.9\times10^{-15}$ & 1551 & 69  & $0.62\pm0.09$ & 0.44 & $ -13^{\rm +11}_{-5}$ & $ 8.5\pm0.1$ &  \ref{B1841-04}\\
B1841$-$05 &    & 0.2557 & $ 9.7\times10^{-15}$ & 3077 & 115  & $0.26\pm0.04$ & 0.26 & & &    \ref{B1841-05}\\
B1842+14 &    & 0.3755 & $ 1.9\times10^{-15}$ & 3073 & 77  & $0.67\pm0.06$ & 0.46 & & &    \ref{B1842+14}\\
B1842$-$04 &    & 0.4868 & $ 1.1\times10^{-14}$ & 1556 & 68  & $0.53\pm0.07$ & 0.45 & & &    \ref{B1842-04}\\
B1844$-$04 & Coh  & 0.5978 & $ 5.2\times10^{-14}$ & 1240 & 115  & $0.38\pm0.03$ & 0.26 & $ 80^{\rm +70}_{-45}$ & $12\pm1$ &  \ref{B1844-04}\\
B1845$-$01 & Dif$^\ast$  & 0.6594 & $ 5.3\times10^{-15}$ & 1031 & 213  & $0.28\pm0.01$ & 0.11 & $ 90^{\rm +170}_{-35}$ & $14\pm2$ &  \ref{B1845-01}\\
J1845$-$0743 &    & 0.1047 & $ 3.7\times10^{-16}$ & 8193 & 115  & $1.02\pm0.09$ & 0.26 & &  &   \ref{J1845-0743}\\
B1846$-$06 &  Lon  & 1.4513 & $ 4.6\times10^{-14}$ & 1208 & 128  & $1.31\pm0.03$ & 0.24 & ?  & $3\pm1$ &  \ref{B1846-06}\\
B1848+13 &    & 0.3456 & $ 1.5\times10^{-15}$ & 1544 & 42  & $0.55\pm0.09$ & 0.55 & &  &   \ref{B1848+13}\\
B1849+00 &    & 2.1802 & $ 9.7\times10^{-14}$ & 1030 & 94  & $0.25\pm0.02$ & 0.17 & &  &   \ref{B1849+00}\\
J1850+0026 &    & 1.0818 & $ 3.6\times10^{-16}$ & 1029 & 41  & $0.61\pm0.08$ & 0.38 &  & &   \ref{J1850+0026}\\
J1852+0305 &    & 1.3261 & $ 1.0\times10^{-16}$ & 1323 & 10  &  & 2.21 & & &    \ref{J1852+0305}\\
J1852$-$2610 &    & 0.3363 & $ 8.8\times10^{-17}$ & 2526 & 42  &  & 0.50 & &  &   \ref{J1852-2610}\\
B1855+02 &    & 0.4158 & $ 4.0\times10^{-14}$ & 2051 & 77  & $0.7\pm0.1$ & 0.49 & & &    \ref{B1855+02}\\
B1855+09 &    & 0.0054 & $ 1.8\times10^{-20}$ & 158193 & 175  & $0.6\pm0.1$ & 0.46 & &  &   \ref{B1855+09}\\
B1857$-$26 & Dif  & 0.6122 & $ 2.0\times10^{-16}$ & 1339 & 360  & $0.47\pm0.02$ & 0.20 & $ 80^{\rm +80}_{-9}$ & $ 7.5\pm0.2$ &  \ref{B1857-26}\\
 &  &    & &  & & &  & $ 180^{\rm +20}_{-45}$ & $ 7.3\pm0.2$ &  \\
B1859+03 &   & 0.6555 & $ 7.5\times10^{-15}$ & 1552 & 182  & $0.35\pm0.05$ & 0.12 & &  &   \ref{B1859+03}\\
B1859+07 &    & 0.6440 & $ 2.3\times10^{-15}$ & 1030 & 46  & $0.47\pm0.08$ & 0.52 & & &    \ref{B1859+07}\\
B1900+01 & Dif$^\ast$  & 0.7293 & $ 4.0\times10^{-15}$ & 1171 & 327  & $0.56\pm0.05$ & 0.09 & $ 30^{\rm +30}_{-3}$ & $ 3.4\pm0.7$ &  \ref{B1900+01}\\
B1900+05 &    & 0.7466 & $ 1.3\times10^{-14}$ & 1025 & 54  & $0.49\pm0.06$ & 0.40 & & &    \ref{B1900+05}\\
J1901$-$0906 & Coh  & 1.7819 & $ 1.6\times10^{-15}$ & 785 & 147  & $0.47\pm0.08$ & 0.13 & $ -11^{\rm +1}_{-8}$ & $ 6.9\pm0.3$ &  \ref{J1901-0906}\\
 &  &    & &  & & &  & $ -5.6^{\rm +0.1}_{-0.7}$ & $ 3.1\pm0.1$ &  \\
B1903+07 &    & 0.6480 & $ 4.9\times10^{-15}$ & 2246 & 32  &  & 0.72 & & &    \ref{B1903+07}\\
B1905+39 &    & 1.2358 & $ 5.4\times10^{-16}$ & 1032 & 50  & $0.78\pm0.07$ & 0.57 & &  &   \ref{B1905+39}\\
B1907+00 &    & 1.0169 & $ 5.5\times10^{-15}$ & 1030 & 40  & $0.63\pm0.06$ & 0.44 & &  &   \ref{B1907+00}\\
B1907+10 &    & 0.2836 & $ 2.6\times10^{-15}$ & 3023 & 95  & $0.51\pm0.07$ & 0.44 & &  &   \ref{B1907+10}\\
B1910+20 &    & 2.2330 & $ 1.0\times10^{-14}$ & 1025 & 43  & $0.67\pm0.07$ & 0.39 & &  &   \ref{B1910+20}\\
B1911+13 &    & 0.5215 & $ 8.0\times10^{-16}$ & 2055 & 56  & $0.32\pm0.04$ & 0.26 & &  &   \ref{B1911+13}\\
B1911$-$04 & Dif$^\ast$  & 0.8259 & $ 4.1\times10^{-15}$ & 2082 & 616  & $0.287\pm0.004$ & 0.08 & $ 70^{\rm +40}_{-30}$ & $15\pm5$ &  \ref{B1911-04}\\
B1914+09 &    & 0.2703 & $ 2.5\times10^{-15}$ & 6145 & 97  & $0.59\pm0.08$ & 0.52 & &  &   \ref{B1914+09}\\
B1914+13 &    & 0.2818 & $ 3.6\times10^{-15}$ & 4096 & 185  & $0.41\pm0.02$ & 0.19 & &  &   \ref{B1914+13}\\
B1915+13 &    & 0.1946 & $ 7.2\times10^{-15}$ & 1541 & 41  &  & 0.46 & & &   \ref{B1915+13}\\
B1916+14 &    & 1.1810 & $ 2.1\times10^{-13}$ & 1226 & 23  &  & 0.79 & & &   \ref{B1916+14}\\
B1917+00 & Dif$^\ast$  & 1.2723 & $ 7.7\times10^{-15}$ & 4106 & 135  & $0.69\pm0.08$ & 0.27 & $ 70^{\rm +7}_{-25}$ & $ 7.8\pm0.2$ &  \ref{B1917+00}\\
B1918+19 &    & 0.8210 & $ 9.0\times10^{-16}$ & 1550 & 60  & $0.58\pm0.07$ & 0.48 & &  &   \ref{B1918+19}\\
B1919+21 & Dif  & 1.3373 & $ 1.3\times10^{-15}$ & 1033 & 207  & $0.30\pm0.04$ & 0.18 & $ -3.4^{\rm +0.3}_{-0.2}$ & $ 4.4\pm0.1$ &  \ref{B1919+21}\\
 &  &    & &  & & &  & $-11^{\rm +1}_{-1}$ & $ 4.4\pm0.1$ &  \\
B1920+21 &    & 1.0779 & $ 8.2\times10^{-15}$ & 1025 & 74  & $0.40\pm0.06$ & 0.30 & &  &   \ref{B1920+21}\\
B1924+16 &   & 0.5798 & $ 1.8\times10^{-14}$ & 2514 & 87  & $0.76\pm0.06$ & 0.44 & &   &    \ref{B1924+16}\\
B1929+10 & Dif$^\ast$  & 0.2265 & $ 1.2\times10^{-15}$ & 1324 & 681  & $0.466\pm0.002$ & 0.04 & $ 90^{\rm +140}_{-8}$ & $ 9.8\pm0.8$ &  \ref{B1929+10}\\
 &  &    & &  & & &  & $ -160^{\rm +10}_{-100}$ & $ 4.4\pm0.1$ &  \\
B1933+16 & Dif$^\ast$  & 0.3587 & $ 6.0\times10^{-15}$ & 2613 & 2081  & $0.240\pm0.001$ & 0.02 & $ 300^{\rm +130}_{-50}$ & $ 2.4\pm0.3$ &  \ref{B1933+16}\\
B1935+25 &    & 0.2010 & $ 6.4\times10^{-16}$ & 7173 & 66  & $0.58\pm0.06$ & 0.47 & & &    \ref{B1935+25}\\
B1937+21 &    & 0.0016 & $ 1.1\times10^{-19}$ & 600160 & 339  &  & 0.19 & & &    \ref{B1937+21}\\
B1937$-$26 &  Lon  & 0.4029 & $ 9.6\times10^{-16}$ & 2129 & 91  & $0.93\pm0.08$ & 0.31 & ? & $ 2.5\pm0.5$ &  \ref{B1937-26}\\
B1943$-$29 &    & 0.9594 & $ 1.5\times10^{-15}$ & 1319 & 43  & $0.8\pm0.1$ & 0.56 & & &    \ref{B1943-29}\\
B1944+17 & Dif$^\ast$  & 0.4406 & $ 2.4\times10^{-17}$ & 2056 & 278  & $1.19\pm0.02$ & 0.17 & $ -11.7^{\rm +0.3}_{-0.3}$ & $ 13\pm5$ &  \ref{B1944+17}\\
B1946+35 &  Lon  & 0.7173 & $ 7.1\times10^{-15}$ & 1161 & 176  & $0.43\pm0.01$ & 0.12 & ? & $33\pm2$ &  \ref{B1946+35}\\
B1952+29 & Dif$^\ast$  & 0.4267 & $ 1.7\times10^{-18}$ & 1536 & 312  & $0.46\pm0.06$ & 0.07 & $ -190^{\rm +15}_{-170}$ & $ 13.1\pm0.2$ &  \ref{B1952+29}\\
 &  &    & &  & & &  & $ -40^{\rm +6}_{-30}$ & $ 12.5\pm0.4$ &  \\
B1953+50 & Dif$^\ast$  & 0.5189 & $ 1.4\times10^{-15}$ & 1635 & 464  & $0.9\pm0.2$ & 0.08 & $ 5.3^{\rm +16}_{-0.2}$ & $ 25\pm7$ &  \ref{B1953+50}\\
B2000+32 &    & 0.6968 & $ 1.1\times10^{-13}$ & 2060 & 90  & $0.8\pm0.1$ & 0.41 & & &    \ref{B2000+32}\\
B2000+40 & Coh  & 0.9051 & $ 1.7\times10^{-15}$ & 768 & 109  & $0.53\pm0.04$ & 0.28 & $ -7.3^{\rm +0.8}_{-0.6}$ & $ 2.5\pm0.1$ &  \ref{B2000+40}\\
 &  &    & &  & & &  & $ -17^{\rm +3}_{-8}$ & $ 2.5\pm0.1$ &  \\
B2002+31 &    & 2.1113 & $ 7.5\times10^{-14}$ & 1115 & 214  & $0.54\pm0.02$ & 0.12 & &  &   \ref{B2002+31}\\
B2003$-$08 &    & 0.5809 & $ 4.6\times10^{-17}$ & 1285 & 37  & $0.8\pm0.2$ & 0.74 & & &    \ref{B2003-08}\\
B2011+38 &  Lon  & 0.2302 & $ 8.9\times10^{-15}$ & 3722 & 271  & $1.34\pm0.02$ & 0.21 & ? & $ 30\pm15$ &  \ref{B2011+38}\\
B2016+28 & Dif$^\ast$  & 0.5580 & $ 1.5\times10^{-16}$ & 25899 & 511  & $0.61\pm0.01$ & 0.15 & $ -70^{\rm +15}_{-25}$ & $ 20\pm2$ &  \ref{B2016+28}\\
 &  &    & &  & & &  & $ -12^{\rm +1}_{-8}$ & $ 4.0\pm0.4$ &  \\
B2020+28 & Dif$^\ast$  & 0.3434 & $ 1.9\times10^{-15}$ & 3759 & 1438  & $0.15\pm0.02$ & 0.04 & $ -55^{\rm +5}_{-15}$ & $ 2.5\pm0.2$ &  \ref{B2020+28}\\
 &  &    & &  & & &  & $ 25^{\rm +15}_{-2}$ & $ 2.3\pm0.1$  &  \\
B2021+51 & Dif$^\ast$  & 0.5292 & $ 3.1\times10^{-15}$ & 20326 & 1180  & $0.50\pm0.03$ & 0.08 & $ 10^{\rm +15}_{-1}$ & $ 5.1\pm0.3$ &  \ref{B2021+51}\\
B2022+50 &    & 0.3726 & $ 2.5\times10^{-15}$ & 2278 & 96  & $0.52\pm0.04$ & 0.31 & & &    \ref{B2022+50}\\
B2043$-$04 & Coh  & 1.5469 & $ 1.5\times10^{-15}$ & 1025 & 122  & $0.69\pm0.02$ & 0.23 & $ 4.5^{\rm +5}_{-0.3}$ & $ 2.74\pm0.05$ &  \ref{B2043-04}\\
B2044+15 & Dif$^\ast$  & 1.1383 & $ 1.8\times10^{-16}$ & 768 & 51  & $0.72\pm0.06$ & 0.40 & $ -7.1^{\rm +0.5}_{-1.4}$ & $18\pm6$ &  \ref{B2044+15}\\
B2045$-$16 & Coh  & 1.9616 & $ 1.1\times10^{-14}$ & 384 & 44  & $0.8\pm0.1$ & 0.55 & $ 17^{\rm +18}_{-2}$ & $ 3.2\pm0.1$ &  \ref{B2045-16}\\
B2053+36 & Dif$^\ast$  & 0.2215 & $ 3.7\times10^{-16}$ & 3594 & 173  & $0.59\pm0.02$ & 0.22 & $ 100^{\rm +340}_{-75}$ & $11\pm2$ &  \ref{B2053+36}\\
B2106+44 &  Lon  & 0.4149 & $ 8.6\times10^{-17}$ & 2067 & 292  & $0.52\pm0.04$ & 0.19 & ? & $ 19\pm7$ &  \ref{B2106+44}\\
B2110+27 & Dif$^\ast$  & 1.2029 & $ 2.6\times10^{-15}$ & 1032 & 370  & $1.06\pm0.02$ & 0.09 & $ 140^{\rm +18}_{-15}$ & $ 4.4\pm0.1$ &  \ref{B2110+27}\\
B2111+46 & Dif  & 1.0147 & $ 7.1\times10^{-16}$ & 3083 & 258  & $0.62\pm0.05$ & 0.17 & $ -180^{\rm +20}_{-50}$ & $ 2.7\pm0.4$ &  \ref{B2111+46}\\
J2145$-$0750 &    & 0.0161 & $ 3.0\times10^{-20}$ & 224536 & 190  & $0.53\pm0.07$ & 0.16 & &  &   \ref{J2145-0750}\\
B2148+52 &    & 0.3322 & $ 1.0\times10^{-14}$ & 2055 & 89  & $0.39\pm0.06$ & 0.34 & &  &   \ref{B2148+52}\\
B2148+63 & Dif$^\ast$  & 0.3801 & $ 1.7\times10^{-16}$ & 2256 & 143  & $0.89\pm0.02$ & 0.26 & $ -35^{\rm +7}_{-15}$ & $ 2.4\pm0.3$ &  \ref{B2148+63}\\
B2154+40 & Dif$^\ast$  & 1.5253 & $ 3.4\times10^{-15}$ & 1148 & 475  & $0.60\pm0.01$ & 0.11 & $ 110^{\rm +80}_{-10}$ & $ 3.1\pm0.8$ &  \ref{B2154+40}\\
B2217+47 &    & 0.5385 & $ 2.8\times10^{-15}$ & 1592 & 191  & $0.52\pm0.03$ & 0.18 & &  &   \ref{B2217+47}\\
B2224+65 &    & 0.6825 & $ 9.7\times10^{-15}$ & 2548 & 56  & $0.96\pm0.07$ & 0.42 & &  &   \ref{B2224+65}\\
B2255+58 & Dif  & 0.3682 & $ 5.8\times10^{-15}$ & 2305 & 506  & $0.52\pm0.01$ & 0.12 & $ 11^{\rm +11}_{-1}$ & $10\pm1$ &  \ref{B2255+58}\\
B2303+30 & Coh  & 1.5759 & $ 2.9\times10^{-15}$ & 1109 & 96  & $0.72\pm0.08$ & 0.26 & $ 15.0^{\rm +3}_{-0.3}$ & $ 2.1\pm0.1$ &  \ref{B2303+30}\\
B2306+55 &    & 0.4751 & $ 2.0\times10^{-16}$ & 3596 & 45  & $0.8\pm0.1$ & 0.70 & & &    \ref{B2306+55}\\
B2310+42 & Coh  & 0.3494 & $ 1.1\times10^{-16}$ & 5006 & 2073  & $0.46\pm0.01$ & 0.11 & $ 60^{\rm +20}_{-10}$ & $ 2.1\pm0.1$ &  \ref{B2310+42}\\
 &  &    & &  & & &  & $ 6.0^{\rm +0.3}_{-0.3}$ & $ 2.1\pm0.1$ &  \\
B2315+21 &    & 1.4447 & $ 1.0\times10^{-15}$ & 1024 & 44  & $0.7\pm0.1$ & 0.58 & & &    \ref{B2315+21}\\
B2319+60 & Coh  & 2.2565 & $ 7.0\times10^{-15}$ & 1041 & 1652  & $0.923\pm0.004$ & 0.03 & $ 70^{\rm +60}_{-10}$ & $ 7.7\pm0.4$ &  \ref{B2319+60}\\
 &  &    & &  & & &  & $ 12^{\rm +4}_{-1}$ & $ 7.7\pm0.1$ &  \\
B2323+63 &    & 1.4363 & $ 2.8\times10^{-15}$ & 1036 & 29  & $1.6\pm0.1$ & 0.67 & & &    \ref{B2323+63}\\
B2324+60 & Dif$^\ast$  & 0.2337 & $ 3.5\times10^{-16}$ & 3633 & 289  & $0.95\pm0.01$ & 0.16 & $ 160^{\rm +40}_{-30}$ & $17\pm4$ &  \ref{B2324+60}\\
B2327$-$20 &    & 1.6436 & $ 4.6\times10^{-15}$ & 1024 & 125  & $0.55\pm0.04$ & 0.15 & &  &   \ref{B2327-20}\\
B2334+61 &    & 0.4953 & $ 1.9\times10^{-13}$ & 3085 & 19  & $1.3\pm0.2$ & 1.07 & &  &   \ref{B2334+61}\\
J2346$-$0609 & Dif$^\ast$  & 1.1815 & $ 1.4\times10^{-15}$ & 770 & 79  & $0.77\pm0.04$ & 0.22 & $ -90^{\rm +25}_{-120}$ & $ 2.3\pm0.1$ &  \ref{J2346-0609}\\
B2351+61 & Dif$^\ast$  & 0.9448 & $ 1.6\times10^{-14}$ & 7655 & 661  & $1.07\pm0.01$ & 0.07 & $ 60^{\rm +25}_{-8}$ & $ 17\pm6$ &  \ref{B2351+61}\\
\hline
\end{tabular}
\end{table}